Author’s response to reviews

Title: Development of an interactive web-based tool to conduct and interrogate meta-analysis of diagnostic test accuracy studies: MetaDTA

Authors:

Suzanne Freeman (suzanne.freeman@leicester.ac.uk)
Clareece Kerby (clareece.kerby@nhs.net)
Amit Patel (ap666@le.ac.uk)
Nicola Cooper (njc21@le.ac.uk)
Terry Quinn (terry.quinn@glasgow.ac.uk)
Alex Sutton (ajs22@le.ac.uk)

Version: 2 Date: 22 Feb 2019

Author’s response to reviews:

1. Please move the Availability and Requirements, Abbreviations and Declarations sections to come between the Conclusions and References.

Response: These sections have been moved as requested.

2. Please remove the Objectives heading from this Abstract, this is not a standard section.

Response: The heading objectives has been removed from the abstract.

3. Figure 1 appears to be a table, please can we suggest uploading Figure 1 as a Table, and changing it to Table 1. Please amend the citations in the text and Figure Legends accordingly.

Response: Figure 1 has been changed to Table 1. All citations in the text and Figure legends have been updated to reflect this change.
4. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (https://emea01.safelinks.protection.outlook.com/?url=http%3A%2F%2Fwww.biomedcentral.com%2Fsubmissions%2Feditorial-policies%23authorship&amp;data=02%7C01%7Csuzanne.freeman%40leicester.ac.uk%7C72d552fbd3fi40c7791c08d693406183%7Caebecd6a31d44b0195ce8274afe853d9%7C0%7C0%7C636858302068778127&amp;sdata=xYDglzanTCF4mYziU32IMJJzB9gfwhIq48Ad3eOCzc%3D&amp;reserved=0).

Currently, the contributions of all authors do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.
If these guidelines are not met, we would request the following change of authorship form be filled out and sent to our editorial office - https://emea01.safelinks.protection.outlook.com/?url=https%3A%2F%2Fresource-cms.springernature.com%2Fspringer-cms%2Frest%2Fv1%2Fcontent%2F7454878%2Fdata%2Fv5&data=02%7C01%7Csuzanne.freeman%40leicester.ac.uk%7C72d552fbd3f40c7791c08d693406183%7Caebecd6a31d44b0195ce8274afe853d9%7C0%7C636858302068778127%7Csdata=hHdjOjdNFAAn6qF5a0f7PpxFNb%2BK0DDdDqujEB0E%3D&reserved=0

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Response: The author contributions section has been re-written to fit in with the journals requirements for this section.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: A single clean version of the manuscript has been uploaded along with separate files for the figures.